Author's response to reviews

Title: Perimenopausal Contraception in Turkish Women: Cross-sectional Study

Authors:

Nevin H Sahin (nevinsahin34@yahoo.com)
Sema B Kharbouch (semaca80@yahoo.com)

Version: 5 Date: 13 December 2006

Author's response to reviews: see over
December 12, 2006

Dear Deborah Saltman, MD PhD

Please find enclosed a copy of the manuscript entitled "Perimenopausal Contraception In Turkish Women: Cross-sectional Study" for consideration as an article in “BMC Nursing” in September 20, 200.

Manuscript has been revised according to comment.

We herewith declare that the manuscript is not submitted to any other journal for review at present and that its language has been reviewed by a professional. The approval and permission of the subjects, who participated in the study, were obtained prior to the initiation of our studies.

Thanking you in advance for your kind efforts we look forward to hearing from you regarding the outcome of the review process.

Sincerely yours

Address:
Nevin Hotun Sahin
Florence Nigtingale Hemsirelik Yuksekokulu
Dogum ve Kadin Hastaliklari Hemsireligi ABD
Sisli 34387, Istanbul, Turkiye
Telephone: +(90) 212 440 00 00/27082,
Fax: +(90) 212 224 4990

E-mail: nevinsah@istanbul.edu.tr
nevinsahin34@yahoo.com